Author's response to reviews

Title: A role for subchondral bone changes in the process of osteoarthritis; a micro-CT study of two canine models.

Authors:

Yvonne H Sniekers (y.sniekers@erasusmc.nl)
Femke Intema (f.intema@umcutrecht.nl)
Floris PJG Lafeber (f.lafeber@umcutrecht.nl)
Gerjo JVM van Osch (g.vanosch@erasusmc.nl)
Johannes PTM van Leeuwen (j.vanleeuwen@erasusmc.nl)
Harrie Weinans (h.weinans@erasusmc.nl)
Simon C Mastbergen (s.mastbergen@umcutrecht.nl)

Version: 3 Date: 6 February 2008

Author's response to reviews: see over
Dear editor

We have made the requested formatting changes to the manuscript (MS1491595161558379), and made a few additional changes, which are listed below.

Yours sincerely,
Yvonne Sniekers

Page 1
Corresponding author is now Simon C Mastbergen instead of Yvonne H Sniekers (indicated with *)

Page 7, 11th line
ccontra-lateral left limb
instead of contra-lateral right limb

Page 7, 18th line
evaluated 10 and 20 weeks
removed ‘,’

Page 16, 4th line
progressive development of OA in the groove model compared to..., added the word ‘model’

Page 16, 6th line
In both models the osteophytes start below the rim
removed ‘at’

Page 17, 18th line
relatively early time points
instead of time-points

Page 19, 3rd line
cartilage-bone interplay is part of the etiology
removed ‘in’